Dear Reviewers,
First of all, we would like to thank you for giving us the opportunity to revise our work. Please find our responses to your comments below.

**Comment:** Flowchart needs to be adjusted and beautified  
**Response:** We have now improved the appearance of flow chart as per your request (Figure 1).

**Comment:** Table 1-4 should be put into the supplemental files, it takes up too much space in the article  
**Response:** We agree with your comments, and we have now put the tables 1-4 into supplemental material.

**Comment:** The reference format is not unified, such as 26, 34 did not provide DOI number.  
**Response:** We added DOI to the papers of our reference list. However, we were unable to identify DOIs for the 26th and 34th reference.

**Comment:** Keywords should be added, such as classification, etc.  
**Response:** Keywords have been added as per your request.

**Comment:** A schematic diagram of the proposed clinical and pathogenetic classification scheme should be added to help readers understand the classification  
**Response:** Please note we have created a diagram to describe our classification proposal as shown in Figure 2.